Author’s response to reviews

Title: Factors associated with Consumption of Alcohol in Older Adults - a Comparison between Two Cultures, China and Norway: The CLHLS and the HUNT-Study

Authors:
Juan Li (lijuanjr@126.com)
Bei Wu (bei.wu@nyu.edu)
Geir Selbæk (geir.selbaek@aldringoghelse.no)
Steinar Krokstad (steinar.krokstad@ntnu.no)
Anne-S Helvik (anne-sofie.helvik@ntnu.no)

Version: 2 Date: 20 Jul 2017

Author’s response to reviews:

Cover letter
July 20, 2017

Editor in Chief
Journal of BMC Geriatrics

Dear Editor,

We would like to thank the editor and the reviewers for their careful consideration and comments of our manuscript, which have been a great help for us to improve our manuscript. We appreciate the opportunity to revise the manuscript entitled “Factors associated with Consumption of Alcohol in Older Adults - a Comparison between Two Cultures, China and Norway: The CLHLS and the HUNT-Study” (BGTC-D-16-00208). We have carefully read the comments and made modifications to the original manuscript. We have re-included the title page, revised the “consent to publish” section, included a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case, and remove
supplementary files. All the authors have read the revision and agreed with the changes. We hope this revision will meet your expectation for publishing.

Please feel free to contact us by email at lijuanjr@126.com if you have any questions regarding this submission.

Thank you for your time and consideration!

Best regards,

Juan Li, MSN (Corresponding author)
Bei Wu, PhD
Geir Selbæk, PhD
Steinar Krokstad, MD
Anne-Sofie Helvik, PhD